% EJC papers *must* begin with the following two lines.
\documentclass[12pt]{article}
\usepackage{e-jc}

% Please remove all other commands that change parameters such as
% margins or pagesizes.

% only use standard LaTeX packages
% only include packages that you actually need

% we recommend these ams packages
\usepackage{amsthm,amsmath,amssymb}

% we recommend the graphicx package for importing figures
\usepackage{graphicx}

% use this command to create hyperlinks (optional and recommended)
\usepackage[colorlinks=true,citecolor=black,linkcolor=black,urlcolor=blue]{hyperref}

% use these commands for typesetting doi and arXiv references in the bibliography
\newcommand{\doi}[1]{\href{http://dx.doi.org/#1}{\texttt{doi:#1}}}
\newcommand{\arxiv}[1]{\href{http://arxiv.org/abs/#1}{\texttt{arXiv:#1}}}


%\usepackage{amsmath,amssymb,latexsym,graphics,epsfig}
%\usepackage{hyperref}
%\usepackage{color}
%\usepackage{amsthm}


% all overfull boxes must be fixed;
% i.e. there must be no text protruding into the margins


% declare theorem-like environments
%%\theoremstyle{plain}
%%\newtheorem{theorem}{Theorem}
%%\newtheorem{lemma}[theorem]{Lemma}
%%\newtheorem{corollary}[theorem]{Corollary}
%%\newtheorem{proposition}[theorem]{Proposition}
%%\newtheorem{fact}[theorem]{Fact}
%%\newtheorem{observation}[theorem]{Observation}
%%\newtheorem{claim}[theorem]{Claim}
%%
%%\theoremstyle{definition}
%%\newtheorem{definition}[theorem]{Definition}
%%\newtheorem{example}[theorem]{Example}
%%\newtheorem{conjecture}[theorem]{Conjecture}
%%\newtheorem{open}[theorem]{Open Problem}
%%\newtheorem{problem}[theorem]{Problem}
%%\newtheorem{question}[theorem]{Question}
%%
%%\theoremstyle{remark}
%%\newtheorem{remark}[theorem]{Remark}
%%\newtheorem{note}[theorem]{Note}


\newtheorem*{theo}{Theorem}
\newtheorem*{lema}{Lemma}
\newtheorem*{propo}{Proposition}

\newtheorem*{ogthm}{Odd-girth theorem}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

% if needed include a line break (\\) at an appropriate place in the title

\title{\bf A short proof of the odd-girth theorem}

% input author, affilliation, address and support information as follows;
% the address should include the country, and does not have to include
% the street address

\author{E.R. van Dam\\
\small Dept. Econometrics and O.R.\\[-0.8ex]
\small Tilburg University\\[-0.8ex]
\small Tilburg, The Netherlands\\
\small\tt edwin.vandam@uvt.nl\\
\and
M.A. Fiol\\
\small Dept. de Matem\`atica Aplicada IV\\[-0.8ex]
\small Universitat Polit\`ecnica de Catalunya\\[-0.8ex]
\small Barcelona, Catalonia\\
\small\tt fiol@ma4.upc.edu
}

%\author{E.R. van Dam$^\ddag$ and M.A. Fiol$^\dag$
%\\ \\
%{\small $^\ddag$Tilburg University, Dept. Econometrics and O.R.} \\
%{\small Tilburg, The Netherlands} {\small (e-mail: {\tt
%edwin.vandam@uvt.nl})} \\
%{\small $^\dag$Universitat Polit\`ecnica de Catalunya, Dept. de Matem\`atica Aplicada IV} \\
%{\small Barcelona, Catalonia} {\small (e-mail: {\tt fiol@ma4.upc.edu})}}
%





% \date{\dateline{submission date}{acceptance date}\\
% \small Mathematics Subject Classifications: comma separated list of
% MSC codes available from http://www.ams.org/mathscinet/freeTools.html}

\date{\dateline{Apr 27, 2012}{July, 2012}\\
\small Mathematics Subject Classifications: 05E30, 05C50}












\def\prova{{\boldmath  $Proof.$}\hskip 0.3truecm}

\def\final{\mbox{ \quad $\Box$}}


\def\ds{\(\displaystyle}
\parindent 0cm
\parskip 2mm

\def\A{\mbox{\boldmath $A$}}
\def\B{\mbox{\boldmath $B$}} \def\Bo{{\cal B}}
\def\C{\mbox{\boldmath $C$}}
\def\Cb{\overline{C}}
\def\Cir{{\cal C}}
\def\Co{\ns{C}}
\def\P{{\cal P}}
\def\Part{{\cal P}}
\def\C{\mathbb C}
\def\N{\mathbb N}
\def\Re{\mathbb R}
\def\Z{\mathbb Z}


\def\diag{\mathop{\rm diag }\nolimits}
%\def\dist{\mathop{\partial }\nolimits}
\def\dist{\mathop{\rm dist }\nolimits}
\def\dgr{\mathop{\rm dgr }\nolimits}
\def\excu{\mbox{$\varepsilon_u$}}
\def\exc{\mbox{$\varepsilon$}}
\def\excD{\mbox{$\varepsilon_{\Dpetita}$}}
\def\exce{\mbox{$\varepsilon_{e}$}}
\def\Ker{\mathop{\rm Ker }\nolimits}
\def\grau{\mbox{\rm deg}}

%%% vectors %%%%%

\def\e{\mbox{\boldmath $e$}}
\def\j{\mbox{\boldmath $j$}}
\def\p{\mbox{\boldmath $p$}}
\def\r{\mbox{\boldmath $r$}}
\def\u{\mbox{\boldmath $u$}}
\def\z{\mbox{\boldmath $z$}}
\def\vecnu{\mbox{\boldmath $\nu$}}
\def\vecrho{\mbox{\boldmath $\rho$}}

\def\vecalpha{\mbox{\boldmath $\alpha$}}
\def\vece{\mbox{\boldmath $e$}}
\def\vecu{\mbox{\boldmath $u$}}
\def\vecv{\mbox{\boldmath $v$}}
\def\vecz{\mbox{\boldmath $z$}}
\def\vecm{\mbox{\boldmath $m$}}
\def\vecj{\mbox{\boldmath $j$}}
\def\vecx{\mbox{\boldmath $x$}}
\def\vecxi{\mbox{\boldmath $\xi$}}

%%% matrius %%%%%

\def\matrixA{\mbox{\boldmath $A$}}
\def\A{\mbox{\boldmath $A$}}

\def\matrixB{\mbox{\boldmath $B$}}

\def\matrixC{\mbox{\boldmath $C$}}

\def\matrixE{\mbox{\boldmath $E$}}
\def\E{\mbox{\boldmath $E$}}
\def\Ei{{\cal E}}

\def\matrixD{\mbox{\boldmath $D$}}
\def\matrixG{\mbox{\boldmath $G$}}
\def\G{\Gamma}

\def\matrixI{\mbox{\boldmath $I$}}
\def\I{\mbox{\boldmath $I$}}

\def\matrixJ{\mbox{\boldmath $J$}}
\def\J{\mbox{\boldmath $J$}}

\def\M{\mbox{\boldmath $M$}}
\def\N{\mbox{\boldmath $N$}}

\def\matrixP{\mbox{\boldmath $P$}}

\def\matrixQ{\mbox{\boldmath $Q$}}
\def\Q{\mbox{\boldmath $Q$}}

\def\matrixR{\mbox{\boldmath $R$}}
\def\R{\mbox{\boldmath $R$}}
\def\S{\mbox{\boldmath $S$}}
\def\matrixU{\mbox{\boldmath $U$}}

%%% espectres %%%%%%%
\def\ev{\mbox{\rm ev}}
\def\sp{\mbox{\rm sp}}
\def\spG{\mbox{\rm sp\,$\Gamma$}}
\def\evG{\mbox{\rm ev\,$\Gamma$}}
\def\spe{\mbox{$\sp_e$\,$\Gamma$}}
\def\eve{\mbox{$\ev_e$\,$\Gamma$}}
%\def\spE{\mbox{\rm sp\,$E$}}
\def\spE{\mbox{$\rm sp_{\Epetita}$\,$\Gamma$}}
%\def\evE{\mbox{\rm ev\,$E$}}
\def\evE{\mbox{$\rm ev_{\Epetita}$\,$\Gamma$}}
\def\evsE{\mbox{$\rm ev_{\Epetita}^{\star}$\,$\Gamma$}}
\def\spi{\mbox{$\rm sp_i$\,$\Gamma$}}
\def\evi{\mbox{$\rm ev_i$\,$\Gamma$}}
\def\spC{\mbox{$\rm sp_{\Cpetita}$\,$\Gamma$}}
\def\spD{\mbox{$\rm sp_{\Dpetita}$\,$\Gamma$}}
\def\spX{\mbox{$\rm sp_{\Xpetita}$\,$\Gamma$}}
\def\evC{\mbox{$\rm ev_{\Cpetita}$\,$\Gamma$}}
\def\evCk{\mbox{$\rm ev_{\Ckpetita}$\,$\Gamma$}}
\def\evCkk{\mbox{$\rm ev_{\Ckkpetita}$\,$\Gamma$}}
\def\evD{\mbox{$\rm ev_{\Dpetita}$\,$\Gamma$}}
\def\evX{\mbox{$\rm ev_{\Xpetita}$\,$\Gamma$}}

\def\qed{\hfill$\Box$}
\def\Cpetita{\mbox{{\tiny $C$}}}
\def\Ckpetita{\mbox{{\tiny $C_k$}}}
\def\Ckkpetita{\mbox{{\tiny $C_k+1$}}}
\def\Dpetita{\mbox{{\tiny $D$}}}
\def\Epetita{\mbox{{\tiny $E$}}}
\def\Npetita{\mbox{{\tiny $N$}}}
\def\subC{\Cpetita}
\def\subD{\Dpetita}
\def\subN{\Npetita}
\def\dsubC{d_{\Cpetita}}
\def\dsubD{d_{\Dpetita}}
\def\dsubX{d_{\Xpetita}}
\def\pbar{\overline{p}}
\newcommand{\co}[1]{\textcolor{blue}{#1}}
\def\alphabar{\overline{\alpha}}
\def\diag{\mathop{\rm diag }\nolimits}
%\def\dist{\mathop{\partial }\nolimits}
%\def\dist{\mathop{\rm dist }\nolimits}
\def\dgr{\mathop{\rm dgr }\nolimits}
\def\ecc{\mathop{\rm ecc }\nolimits}
\def\mod{\mathop{\rm mod }\nolimits}
\def\pbar{\overline{p}}
\def\Ker{\mathop{\rm Ker }\nolimits}
\def\tr{\mathop{\rm tr }\nolimits}
\def\som{\mathop{\rm sum }\nolimits}
\def\ev{\mathop{\rm ev }\nolimits}
\def\sp{\mathop{\rm sp }\nolimits}
\def\span{\mathop{\rm span }\nolimits}
\def\sup{\mathop{\rm sup }\nolimits}


\begin{document}
\begin{thebibliography}{99}

\bibitem{b93} N. Biggs, \emph{Algebraic Graph Theory}, Cambridge University
    Press, Cambridge, 1974, second edition, 1993.

\bibitem{bcn89}
A.E. Brouwer, A.M. Cohen, and A. Neumaier, \emph{Distance-Regular Graphs},
Springer-Verlag, Berlin-New York, 1989.

\bibitem{bh12} A.E. Brouwer and W.H. Haemers, \emph{Spectra of Graphs},
    Springer,
    2012; available online at
    \url{http://homepages.cwi.nl/~aeb/math/ipm/}.


\bibitem{cds82} D.M. Cvetkovi\'c, M. Doob, and H. Sachs, \emph{Spectra of
    Graphs}, VEB Deutscher Verlag der Wissenschaften, Berlin,
    second edition, 1982.

\bibitem{vd08} E.R. van Dam, The spectral excess theorem for distance-regular
    graphs: a global (over)view, {\em Electron. J. Combin.} 15(1) (2008),
    \#R129.


\bibitem{vdh11} E.R. van Dam and W.H. Haemers, An odd characterization of the
    generalized odd graphs, {\em J. Combin. Theory Ser. B} 101 (2011),
    486--489.

\bibitem{dkt12} E.R. van Dam, J.H. Koolen, and H. Tanaka, Distance-regular
    graphs,
    manuscript (2012), available online at
    \url{http://lyrawww.uvt.nl/~evandam/files/drg.pdf}.

\bibitem{f02} M.A. Fiol, Algebraic characterizations of distance-regular
    graphs, \emph{Discrete Math.} 246 (2002), 111--129.


\bibitem{fgg10} M.A. Fiol, S. Gago, and E. Garriga, A simple proof of the
    spectral excess theorem for distance-regular graphs, {\em Linear Algebra
    Appl.} 432 (2010), 2418--2422.

\bibitem{fg97} M.A. Fiol and E. Garriga, From local adjacency polynomials to
    locally pseudo-distance-regular graphs, \emph{J. Combin. Theory Ser. B} 71 (1997), 162--183.

\bibitem{fgy96b} M.A. Fiol, E. Garriga, and J.L.A. Yebra, Locally
    pseudo-distance-regular graphs, {\em J. Combin. Theory Ser. B} 68
    (1996), 179--205.


\bibitem{gmk80} C.D. Godsil and B.D. McKay, Feasibility conditions for the
    existence of walk-regular graphs, {\em Linear Algebra Appl.} 30 (1980),
    51--61.

\bibitem{HL99} T. Huang and C. Liu, Spectral characterization
    of some generalized odd graphs. {\it Graphs Combin.} 15
    (1999), 195--209.

\bibitem{lw11} G.-S. Lee and C.-w. Weng, The spectral excess theorem for
    general
    graphs, {\em J. Combin. Theory Ser. A} 119 (2012), 1427--1431.

\end{thebibliography}


\maketitle

\begin{abstract}
\noindent Recently, it has been shown that a connected graph $\G$ with $d+1$
distinct eigenvalues and odd-girth $2d+1$ is distance-regular. The proof of
this result was based on the spectral excess theorem. In this note we present
an alternative and more direct proof which does not rely on the spectral excess
theorem, but on a known characterization of distance-regular graphs in terms of
the predistance polynomial of degree $d$.
\end{abstract}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Introduction}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
The spectral excess theorem \cite{fg97} states that a connected regular graph
$\G$ is distance-regular if and only if its spectral excess (a number which can
be computed from the spectrum of $\G$) equals its average excess (the mean of
the numbers of vertices at maximum distance from every vertex), see
\cite{vd08,fgg10} for short proofs. Using this theorem, Van Dam and Haemers
\cite{vdh11} proved the below odd-girth theorem for regular graphs.

\begin{ogthm} A connected graph with $d+1$ distinct
eigenvalues and finite odd-girth at least $2d+1$ is a distance-regular
generalized odd graph $($that is, a distance-regular graph with diameter $D$
and odd-girth $2D+1$\/$)$. \end{ogthm}

In the same paper, the authors posed the problem of deciding whether the
regularity condition is necessary or, equivalently, whether or not there are
nonregular graphs with $d+1$ distinct eigenvalues and odd girth $2d+1$.
Moreover, they proved this in the negative for the case $d+1=3$, and claimed to
have proofs for the cases $d+1\in \{4,5\}$. In a recent paper, Lee and Weng
\cite{lw11} used a variation of the spectral excess theorem for nonregular
graphs to show that, indeed, the regularity condition is not necessary. The
odd-girth theorem generalizes the result by Huang and Liu \cite{HL99} that
states that every graph with the same spectrum as a generalized odd graph must
be such a graph itself. Well-known examples of generalized odd graphs are the
odd graphs and the folded cubes.

In this note we give a short and direct proof of the more general result
without using any of the spectral excess theorems, but only a known
characterization of distance-regularity in terms of the predistance polynomial
$p_d$ of highest degree.



\section{Preliminaries}
Here we give some basic notation and results on which our proof of the
odd-girth theorem is based. For more background on spectra  of graphs,
distance-regular graphs, and their characterizations, see
\cite{b93,bcn89,bh12,cds82,dkt12,f02}.

Let $\G$ be a connected graph with vertex set $V$, order $n=|V|$, and adjacency
matrix $\A$. The spectrum of $\G$ (that is, of $\A$) is denoted by $\sp \G =
\{\lambda_0^{m_0},\lambda_1^{m_1},\dots, \lambda_d^{m_d}\}$, with distinct
eigenvalues $\lambda_0>\lambda_1>\cdots >\lambda_d$, and corresponding
multiplicities $m_i=m(\lambda_i)$. The {\em predistance polynomials} $p_i$
($i=0,1,\ldots,d$) of $\G$, form a sequence of orthogonal polynomials with
respect to the scalar product $\langle f,g\rangle=\frac{1}{n}\sum_{i=0}^d m_i
f(\lambda_i)g(\lambda_i)$, normalized in such a way that
$\|p_i\|^2=p_i(\lambda_0)$. Then, modulo the minimal polynomial of $\A$, these
polynomials satisfy a three-term recurrence relation of the form
\begin{equation}
\label{recur-pol}
xp_i=\beta_{i-1}p_{i-1}+\alpha_i p_i+\gamma_{i+1}p_{i+1}\qquad
(i=0,1,\dots,d),
\end{equation}
where we let $\beta_{-1}p_{-1}=0$ and $\gamma_{d+1}p_{d+1}=0$. Note that if
$\G$ is distance-regular and $\A_i$ stands for the distance-$i$ matrix, then
$\A_i=p_i(\A)$ for $i=0,1,\ldots,d$. Our proof of the odd-girth theorem relies
mainly on the following result which was first proved in \cite{fgy96b} (see
also \cite{vd08}, \cite{fgg10}).

\begin{propo}
\label{propo-charac-drg} A connected regular graph $\G$ with $d+1$ distinct
eigenvalues is distance-regular if and only if the predistance polynomial $p_d$
satisfies $p_d(\A)=\A_d$.
\end{propo}

We remark that it is fairly easy to prove this characterization by backward
induction, using the recurrence relation \eqref{recur-pol}, the fact that the
Hoffman polynomial $H$ equals $p_0+p_1+\cdots+p_d$, and that for regular graphs
$H(\A)$ equals the all-1 matrix $\J$.

By the matrices  $\E_i$ we denote the {\it $($principal\/$)$ idempotents} of
$\A$ representing the orthogonal projections of $\Re^n$ onto the eigenspaces
$\Ei_i=\Ker (\A-\lambda_i \I)$, for $i=0,1,\dots,d$. In particular, if $\G$ is
regular, then the all-$1$ vector $\j$ is a $\lambda_0$-eigenvector and
$\E_0=\frac{1}{n}\j\j^{\top}=\frac{1}{n}\J$. The diagonal entries of these
idempotents, $m_{u}(\lambda_i)=(\E_i)_{uu}$, have been called the {\em
$u$-local multiplicities} of the eigenvalue $\lambda_i$. Graphs for which these
local multiplicities are independent of the vertex $u$ (that is, for which
every idempotent has constant diagonal) are called {\em spectrum-regular}. The
local multiplicities allow us to compute the number of closed $\ell$-walks from
$u$ to itself in the following way:
\begin{equation}
\label{crossed-mul->num-walks} a_{u}^{({\ell})} =(\A^{\ell})_{uu}=
\sum_{i=0}^d m_{u}(\lambda_i)\lambda_i^{\ell} \qquad (\ell =0,1,2,\ldots).
\end{equation}

A graph is {\em walk-regular} (a concept introduced by Godsil and McKay
\cite{gmk80}) if the number $a_{u}^{({\ell})}$ of closed walks of length $\ell$
does not depend on $u$, for every $\ell=0,1,2,\ldots$. Clearly, a graph is
walk-regular if and only if it is spectrum-regular, and every walk-regular
graph is regular; properties that will be used in our proof of the odd-girth
theorem.



\section{The proof}

Now let us consider a connected graph $\G$ with $d+1$ distinct eigenvalues and
finite odd-girth (at least) $2d+1$, and the corresponding predistance
polynomials with recurrence \eqref{recur-pol}. As was shown in \cite{vdh11} by
an easy inductive argument, in this particular case we have that $\alpha_i=0$
for $i=0,1,\ldots,d-1$ and the polynomials $p_i$ are even or odd depending on
$i$ being even or odd, respectively. Moreover, $\alpha_d \neq 0$ (even though
Van Dam and Haemers \cite{vdh11} restrict to regular graphs, the regularity
condition is not used by them; the argument is also implicitly used by Lee and
Weng \cite{lw11}). In order to prove the odd-girth theorem, we first need the
following lemma.

\begin{lema}
\label{plusminuslemma}
Let $\G$ be a connected graph with $d+1$ distinct
eigenvalues and odd-girth $2d+1$. If $\lambda$ is an eigenvalue of $\G$, then
$-\lambda$ is not. In particular, all eigenvalues are nonzero.
\end{lema}
%\prova
\begin{proof} Assume that both $\lambda$ and $-\lambda$ are eigenvalues
of $\G$, that is, they are both roots of the minimal polynomial. This means
that we can plug in $\lambda$ and $-\lambda$ in the recurrence relations
\eqref{recur-pol} and, in particular, we obtain the two equations $\pm \lambda
p_d(\pm \lambda)=\beta_{d-1}p_{d-1}(\pm \lambda)+\alpha_dp_d(\pm \lambda)$. By
using that the predistance polynomials are odd or even as indexed, and that
$\alpha_d \neq 0$, it follows that $p_d(\lambda)=0$ (also in the case that
$\lambda=0$), which is a contradiction (because by the recurrence relations
this would imply that $p_i(\lambda)=0$ for all $i$, including $i=0$, but
$p_0=1$).  \end{proof}%\final

Now we are ready to prove the general setting of the odd-girth theorem without
using the spectral excess theorem.

\begin{theo}
\label{main-theo}
A connected graph $\G$ with $d+1$ distinct eigenvalues and odd-girth $2d+1$ is distance-regular.
\end{theo}
%\prova
\begin{proof} Fist, let us prove that $\G$ is spectrum-regular (or walk-regular).
Since the number of odd cycles with length at most $2d-1$ is zero we have,
using \eqref{crossed-mul->num-walks},
\begin{equation*}
\sum_{i=1}^d m_u(\lambda_i)\lambda_i^{2\ell-1}=-m_u(\lambda_0)\lambda_0^{2\ell-1} \qquad (\ell=1,2,\ldots,d).
\end{equation*}
This can be seen as a determined system of $d$ equations and $d$ unknowns
$m_u(\lambda_i)$ ($i=1,2,\ldots,d$). Indeed, by the properties of Vandermonde
matrices and the above lemma, the determinant of its coefficient matrix is
\begin{eqnarray*}
\left|
\begin{array}{cccc}
\lambda_1  & \lambda_2 & \cdots & \lambda_d \\
\lambda_1^3  & \lambda_2^3 & \cdots & \lambda_d^3 \\
\vdots &\vdots & \ddots &  \vdots \\
\lambda_1^{2d-1}  & \lambda_2^{2d-1} &\cdots & \lambda_d^{2d-1}
\end{array}
\right| & =& \prod_{i=1}^d\lambda_i\left|\begin{array}{cccc}
1  & 1 & \cdots & 1 \\
\lambda_1^2  & \lambda_2^2 & \cdots & \lambda_d^2 \\
\vdots &\vdots & \ddots &  \vdots \\
\lambda_1^{2d-2}  & \lambda_2^{2d-2} &\cdots & \lambda_d^{2d-2}
\end{array}\right| \\
 & = & \prod_{i=1}^d\lambda_i\prod_{d\ge i>j\ge 1}(\lambda_i^2-\lambda_j^2) \neq 0.
\end{eqnarray*}
Thus, there exist constants $\alpha_i$ such that $m_u(\lambda_i)=\alpha_i
m_u(\lambda_0)$, for $i=0,1,\ldots,d$. From this it follows that
$m_u(\lambda_0)\sum_{j=0}^d \alpha_j=\sum_{j=0}^dm_u(\lambda_j)=1$, where the
last equality follows from the fact that the sum of all idempotents equals the
identity matrix. Thus, for every $i=0,1,\ldots,d$,
$m_u(\lambda_i)=\alpha_i/\sum_{j=0}^d \alpha_j$, which does not depend on $u$,
and hence $\G$ is spectrum-regular (and walk-regular).

Next, let us show that $p_d(\A)=\A_d$. Since $\G$ is regular, the Hoffman
polynomial $H=p_0+p_1+\cdots+p_d$ satisfies $H(\A)=\J$ and hence
$(p_d(\A))_{uv}=1$ if $\dist(u,v)=d$. Besides, from the parity of the
predistance polynomials, it follows that $(p_i(\A))_{uv}=0$ if $\dist(u,v)$ and
$i$ have different parity (otherwise, $\G$ would have an odd cycle of length
smaller than $2d+1$). So $(p_d(\A))_{uv}=0$ for every pair of vertices $u,v$
whose distance has a different parity than $d$. If $\dist(u,v)$ is smaller than
$d$, but with the same parity, then from the recurrence \eqref{recur-pol} we
get
$$
(\A p_d(\A))_{uv}=\beta_{d-1}(p_{d-1}(\A))_{uv}+\alpha_d (p_d(\A))_{uv}=
\alpha_d (p_d(\A))_{uv}
$$
(because $\dist(u,v)$ and $d-1$ have different parity). But the first term is
$$
\sum_{w\in V} (\A)_{uw} (p_d(\A))_{wv}=\sum_{w\in \G(u)}(p_d(\A))_{wv}=0
$$
since $\dist(w,v)=\dist(u,v)\pm 1$ has a different parity than $d$. Thus, as
$\alpha_d\neq 0$, we find that also in this case $(p_d(\A))_{uv}=0$.
Consequently, $p_d(\A)=\A_d$ and by the above proposition, $\G$ is
distance-regular. \end{proof}%\final


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%Bibliografia
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%






\end{document}
